Author’s response to reviews

Title: Midwives' Experiences of Performing Maternal Observations and Escalating Concerns: a Focus Group Study

Authors:

Justine Jeffery (Justine.Jeffery@heartofengland.nhs.uk)
Alistair Hewison (a.hewison@bham.ac.uk)
Laura Goodwin (L.Goodwin@bham.ac.uk)
Sara Kenyon (s.kenyon@bham.ac.uk)

Version: 2 Date: 26 Jul 2017

Author’s response to reviews:

Dear Dr van Teijlingen,

Many thanks for your positive review of our manuscript, along with your constructive feedback. I have addressed the two style issues mentioned in your revision request:

- The opening paragraph in the abstract has been changed to reflect that the maternal mortality reports are from the United Kingdom.

- On page 6, lines 30-31, the use of ‘yrs' has been changed to 'years' to maintain consistency with the rest of the manuscript.

Both of these revisions can be seen as tracked changes in the uploaded manuscript.

Best wishes,

Laura Goodwin